Author's response to reviews

Title: Longitudinal fluctuations in PD1 and PD-L1 expression in association with changes in anti-viral immune response in chronic hepatitis B

Authors:

Zhang Wenjin (yinzhou@zju.edu.cn)
Peng Chuanhui (yuxb@zju.edu.cn)
Wan Yunle (wanyldr@163.com)
Zheng Shusen (xiehaiyang@yeah.net)

Version: 2 Date: 26 May 2012

Author's response to reviews:

longitudinal fluctuations in PD1 and PD-L1 expression in association with changes in anti-viral immune response in chronic hepatitis B

Zhang Wenjin1, Peng Chuanhui1, Wan Yunle2, Zheng Shusen1*

1 Key Laboratory of Combined Multi-organ Transplantation, Ministry of Public Health, First Affiliated Hospital, School of Medicine, Zhejiang University, Hangzhou, China

2 Department of Hepatobiliary Surgery, Sun Yat-Sen Memorial Hospital, Sun Yat-Sen University, Guangzhou 510120, Guangdong Province, China

*Corresponding Author: Zheng Shusen, MD, PhD, FACS, Key Laboratory of Combined Multi-organ Transplantation, Ministry of Public Health, First Affiliated Hospital, School of Medicine, Zhejiang University, 79 Qingchun Road, Hangzhou 310003, China

Tel: 86-571-87236466
Fax: 86-571-87236884
Email: yinzhou@zju.edu.cn

Running title: Longitudinal fluctuations in PD1 and PD-L1 expression in chronic hepatitis B